Author's response to reviews

Title: Anastrozole-related acute hepatitis with autoimmune features: a case report.

Authors:

Alessandro Inno (alessandro.inno@gmail.com)
Michele Basso (michele.basso@rm.unicatt.it)
Fabio M Vecchio (fmvecchio@rm.unicatt.it)
Valentina A Marsico (valentina.marsico@gmail.com)
Eleonora Cerchiaro (eleonoracer@hotmail.com)
Ettore D'Argento (edargento@libero.it)
Cinzia Bagalà (cinziabagala@libero.it)
Carlo Barone (carlobarone@rm.unicatt.it)

Version: 4 Date: 15 March 2011

Author's response to reviews: see over
Dear Editor,

I am very pleased that the manuscript has been accepted for publication on your journal. In present version I have made the formatting changes you required.

In detail:

**Major revisions:**

a) Author list (...) 
I have modified the list according to the preferred style.

b) Figures (...) 
The figure has been re-drawn as a single composite illustration and built in a single image file.

c) References (...) 
References have been cited according your suggestions, with consecutive numbers in square brackets and before the full stop.

**Minor revisions:**

Abstract and Introduction
The term “Introduction” has been changed with “Background”, both in the abstract and in the manuscript body.
Highlighting/tracking
All tracking and highlighting have been removed.

Typography
The manuscript has been carefully checked for typographical errors.

Yours sincerely,

Prof. Carlo Barone, MD, PhD

Carlo Barone, MD, PhD
Oncologia Medica
Università Cattolica del S. Cuore
Largo F. Vito, 1
00168 Roma - ITALY
Tel: 0039 06 30156318
Fax: 0039 06 30154838
E-mail: carlobarone@rm.unicatt.it